Author’s response to reviews

Title: Eliciting meta consent for future secondary research use of health data using a smartphone application - A proof of concept study in the Danish population

Authors:

Thomas Ploug (ploug@hum.aau.dk)

Søren Holm (soren.holm@manchester.auc.uk)

Version: 4 Date: 31 Jul 2017

Author’s response to reviews:

Dear Editor(s)

We have added two lines spelling it out that we are still analyzing data but that they can be made available on reasonable request.

Yours,

Thomas Ploug